Open Access



Effects of depot medroxyprogesterone acetate, the copper IUD and the levonorgestrel implant on testosterone, sex hormone binding globulin and free testosterone levels: ancillary study of the ECHO randomized clinical trial

G. Justus Hofmeyr^{1,2,3,4*}[®], Mandisa Singata-Madliki^{5,6}[®], Joanne Batting^{2,5}, Yusentha Balakrishna⁶[®] and Chelsea Morroni^{7,8}[®]

Abstract

Background Robust information on relative effects of hormonal contraceptives on endogenous androgens is important for understanding beneficial and adverse effects, method choice and development of new methods.

Methods In this ancillary study at the East London, South Africa site of the ECHO multicentre randomized trial, we compared effects of three contraceptive methods on serum androgen levels among contraceptive users aged 18 to 35 years. Participants were allocated by centrally-managed randomization to open label depot medroxyprogester-one acetate (DMPA-IM), copper intrauterine device (IUD) or levonorgestrel implant.

The primary outcome was free testosterone at 6 months.

Results We analysed stored baseline and 6-month serum samples in 398/615 participants (DMPA-IM 131/205, IUD 135/205 and implant 132/205). Median testosterone levels at baseline were DMPA-IM 0.82, IUD 0.9 and implant 0.87 nmol/L; at 6 months, DMPA 0.68 (lower than IUD, mean percentage difference 28.35, (p < 0.001), IUD 0.86 (unchanged) and implant 0.66, lower than IUD, mean percentage difference – 22.98, p < 0.001).

Median SHBG levels at baseline were DMPA 52.4, IUD 50.5 and implant 55.75 nmol/L; at 6 months, DMPA 40.65, lower than IUD (mean percentage difference 21.19, p = 0.005), IUD 49.1 (unchanged), and implant 23.35 nmol/L, lower than IUD (mean percentage difference – 50.04, p < 0.001 and than DMPA (mean percentage difference – 39.45, p < 0.001).

The Evidence for Contraceptive options and HIV Outcomes (ECHO) Trial was a multi-center, open label, randomized clinical trial comparing HIV incidence and contraceptive benefits in women using depot medroxyprogesterone acetate (DMPA-IM), levonorgestrel (LNG) implants, and copper intrauterine devices (IUD). FHI 360 Study #523201

*Correspondence: G. Justus Hofmeyr justhof@gmail.com Full list of author information is available at the end of the article



© The Author(s) 2024. **Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, wish http://creativecommons.gr/licenses/by/4.0/. The Creative Commons Public Domain Dedication waiver (http://creativecommons.gr/licenses/by/4.0/. The Creative Commons Public Domain Dedicated in a credit line to the data.

Free testosterone levels at baseline were DMPA 10, IUD 12 and implant 11 pmol/L; at 6 months, DMPA 11, less than IUD (mean percentage difference 13.53, p = 0.047), IUD 12 and implant 14, higher than IUD (mean percentage difference 14.15, p = 0.038) and than DMPA, (mean percentage difference 29.60, p < 0.001).

Conclusions This is the first randomized trial to show lower SHBG and higher free testosterone with the levonorgestrel implant than with DMPA, and contrasts with reports of increased SHBG with combined oral ethinyl estradiol/ levonorgestrel use, and reduced androgens (and impaired sexual function) reported with the etonorgestrel implant. The higher free testosterone with the LNG implant might improve sexual function, mood and bone health as well as increasing side-effects such as acne and hirsutism, and is consistent with the greater sexual activity (with respect to multiple sex partners, new sex partner and unprotected sex) with the implant compared with DMPA documented in the ECHO study.

ECHO trial registration Clinical Trials.gov, number NCT02550067 15/09/2015.

Plain English summary Contraception, or family planning, is central to the role of women in societies. It is most important to have accurate information on the relative side-effects of various contraceptive options in order to empower women to make informed choices regarding their preferred method.

Hormonal contraceptives contain various forms of the female sex hormones, estrogens and/or progestogens. These hormones have direct effects on the users, as well as modifying the levels of the users' own circulating sex hormones, both the 'female' and the 'male' sex hormones (androgens).

In this study, consenting participants requesting contraception, were allocated randomly to receive either depot medroxyprogesterone acetate (DMPA-IM) a 3-monthly progestogen injection, the copper intrauterine device (IUD), a non-hormonal contraceptive inserted within the womb, or the levonorgestrel implant, a device placed under the skin which releases a progestogen for 5 years.

We measured the participants' androgen levels after 6 months, and found for the first time that the active form of testosterone (free testosterone) was 29% higher with the implant than with DMPA-IM. The level with the IUD was intermediate, and significantly different from the other two methods.

This finding is relevant to the effects experienced by users of these methods, because free testosterone has effects on sexual function, bone health and mood, as well as on conditions such as acne and hair distribution patterns.

Keywords Contraception, Depot medroxyprogesterone acetate, Copper intrauterine device, Levonorgestrel implant, Testosterone, Sex hormone binding globulin, Free testosterone, Randomized trial

Tweetable abstract

Higher free testosterone with LNG implant than DMPA intramuscular contraception may impact acceptance and sexual behaviour.

Background

Hormonal contraception has complex effects on users' endocrine systems. Apart from direct pharmacological effects of the exogenous contraceptive hormones, endogenous sex steroid hormone levels are altered. It is important to have accurate data on such effects, and in particular differential effects of alternative contraceptive methods, in order to be able to counsel users on the relative benefits and risks of alternative methods, to understand the potential clinical impact of these effects, and to guide future developments of contraceptive methods.

Previous consideration of androgenic effects of contraceptives such as oily skin, acne [1], hirsutism, android obesity, androgenic alopecia, unfavourable

lipid profiles, diabetes and hypertension, has tended to focus on the direct androgenic effects of the exogenous progestins [2] rather than secondary effects on endogenous androgens. Minimizing androgenic side-effects is considered a key factor in the acceptability and continuation of hormonal contraception [3].

In addition to multiple physiological effects, sex steroid hormones have important neuropsychological and behavioural effects which are complex and poorly understood. Oral contraceptive discontinuation for depression or loss of libido appears to be less common with lower dose than the older high-dose formulations [4]. Low androgen levels have been associated with increased pelvic pain, dysmenorrhoea and headache [5]. Androgens are thought to increase libido, though results of previous studies are conflicting. Sexual function may be impaired by certain oral contraceptives and restored by androgen replacement [6]. Estrogens stimulate hepatic sex hormone binding globulin (SHBG) production [7]. Oral contraception with ethinyl estradiol combined with either levonorgestrel (LNG) or drospirenone increased SHBG and reduced testosterone and free testosterone levels. Addition of dehydroepiandrosterone (50 mg/day orally) normalized free testosterone [8]. Placebo-controlled randomized trials of testosterone treatment have shown improvement in hypoactive sexual desire disorder [7]. Testosterone has also been associated with sense of control and dominance.

Hormone-related effects on sexual behaviour and changes in sexual exposure during menstruation may affect susceptibility to sexually transmitted infections [9, 10].

In a previous randomized trial, we found reduced sexual activity among participants randomized to injectable progestogens versus the copper intrauterine device (IUD) [11, 12]. In the ECHO trial, reduced sexual activity and reduced condomless sexual activity was reported by participants allocated to depot medroxyprogesterone acetate intramuscular (DMPA-IM) compared with both the levonorgestrel implant and the IUD [13].

Previous research has found that some progestogenic contraceptives suppress testosterone levels. Subcutaneous DMPA 104 mg 12-weekly was associated with decreased total testosterone and sex hormone binding globulin (SHBG) to 26 weeks [14]. Lower levels of free testosterone were found among combined oral contraceptive than DMPA users [15]. Etonogestrel implants were associated with impaired sexual function, ascribed to suppressed androgen and estrogen levels [16].

Observational studies are intrinsically subject to confounding. The ECHO study presents a unique opportunity to compare testosterone and sex hormone binding globulin (SHBG) levels between participants allocated to DMPA-IM, the copper IUD and the levonorgestrel (LNG) implant in the context of a rigorous randomized clinical trial.

Objectives

We compared testosterone, SHBG and free testosterone levels between women randomly allocated DMPA IM, the copper IUD or the LNG implant in order identify differences which might be associated with clinically important side effect profiles.

Methods

This is an ancillary study of the ECHO study, limited to participants enrolled at the Effective Care Research Unit site in East London, South Africa between 1 March 2016 and 14 August 2017. The ECHO study protocol [17] and primary paper [13] have been published previously. Briefly, HIV uninfected participants requesting contraception aged 18 to 35 years who indicated that they had

not used injectable hormonal contraception in the preceding 6 months, were allocated in parallel, in balanced blocks of 15-30, stratified by site, using an online randomization service, in 1:1:1 ratio to receive DMPA-IM 3-monthly, the copper T 380A IUD or the LNG implant, and were followed after 1 month then 3-monthly for 12 to 18 months. All participants gave written informed consent to participate. Participants were enrolled by research staff, and there was no blinding. Blood samples were collected at baseline and at the 3-monthly visits, separated on site and the serum stored at -80°C in the BARC-SA Bio Repository in Johannesburg. We chose the 6-month interval to allow adequate time for stabilisation of the effects of the hormonal contraceptives. For this ancillary study, a convenience sample size of all available specimens was used.

Laboratory analysis

Laboratory tests were conducted by BARC-SA (Pty) Ltd. Testosterone and SHBG were measured at baseline and closest available sample to 6 months by chemiluminescent microparticle one-step immunoassay for quantitative measurement (CMIA) using the Abbott Architect analyser (sensitivity 0.05 and 0.02 nmol/L respectively; testosterone potential interference <10%; SHBG: no detectable cross-reactivity). Free testosterone was calculated according to the method of Vermeulen et al. [18] The few results above or below the detection level of the assay were assigned the proxy value of the upper or lower limit of the assay respectively.

Data analysis

The laboratory data were entered into an Excel spreadsheet, cleaned and merged with data from the original ECHO study database for analysis. Baseline data were compared to ensure that the comparability of the groups was not compromised by loss to follow-up, which was similar between groups. Core outcomes were not relevant to this hormonal study.

Baseline categorical demographic variables were compared between arms using Pearson's chi-square or Fisher's exact test, where applicable. Baseline continuous demographic variables were compared between arms using ANOVA. Normality was tested using the Shapiro-Wilk test. Since data was non-normally distributed, medians (with interquartile ranges) were reported for baseline, six-month and change from baseline values. Differences between baseline and six-month values for each allocated method were tested using the Wilcoxon signed-rank test to account for the paired data. To facilitate pairwise comparisons between the allocated methods, the data was log-transformed (natural log) and a mixed-effects linear regression model was fitted for each of the outcomes (testosterone, SHBG, free testosterone). The allocated method, time point and the interaction between allocated method and time point were included as fixed effects. The participant was included as a random effect to account for the within-subject correlation between repeated measures. The distribution of the residuals was visually assessed using quantile-quantile plots. The mean differences between allocated arms are presented as percentages with 95% confidence intervals (CIs). Results were considered significant for p < 0.05. It was not considered useful to adjust for daily variation in hormone levels [19] because these would be balanced out by the large sample size in each group.

Patient and public involvement

Extensive patient and public involvement is documented in the primary ECHO study paper [13]. Community stakeholders were actively involved in the protocol design. Community advisory groups at each site provided input into conduct of the trial on an ongoing basis.

Results

Of 615 participants enrolled at the site, results were available for 398 (65%) participants both at baseline and at 6 months (Fig. 1).

Screened

Enrolment

Baseline Demographic data were similar between groups (Table 1).

In the main trial, random testing of baseline blood of 157/615 participants for evidence of recent hormonal contraceptive exposure found quantifiable medroxy-progesterone acetate in 85 (54%), levonorgesterel in 9 (5.7%), norethisterone in 7 (4.5%) and etonogestrel in 1 (0.6%). Given the potential for distortion of baseline hormone levels due to prior contraceptive exposure, we have primarily compared absolute differences between randomized groups at 6 months as well as change from baseline. Prior exposure to progestogen contraception would be expected to be similar between randomly allocated groups, and would result in under-estimation of the changes observed in the progestogen groups. Hormonal results are shown in Tables 2 and 3.

Testosterone

Assessed for eligibility (n= 889)

Randomized (n= 615)

Allocation

Excluded (n= 274)

Not meeting inclusion criteria (n= 262) Did not complete all screening procedures (n= 4) Declined randomization (n= 8)

Baseline median (interquartile range) testosterone levels were similar between the three groups (DMPA 0.82 (0.56, 1.2), IUD 0.9 (0.66, 1.18), implant 0.87 (0.64, 1.26) nmol/L). At 6 months, median testosterone levels were significantly reduced for DMPA (0.68 (0.5, 0.88)) and the levonorgestrel implant (0.66 (0.51, 0.85) nmol/L), which were similar, and both were significantly lower



Fig. 1 CONSORT flow diagram. *n = 2 did not receive the allocated intervention

	DMPA			IUD			Implant			р
	n			n			n			
Age (years)	121	25.0	(4.3)	124	25.4	(4.6)	120	24.5	(4.7)	0.30
Never married	131	122	93.1%	135	131	97.0%	136	132	97.1%	0.23
Secondary school incomplete	131	75	57.3%	135	63	46.7%	136	76	55.9%	0.17
Earns income	131	16	12.2%	135	22	16.3%	136	19	14.0%	0.63
Nulliparous	131	33	25.2%	135	40	29.6%	136	41	30.2%	0.62
Regular menses	131	76	58.0%	135	89	65.0%	136	80	58.8%	0.34
Breastfeeding	131	19	14.5%	135	12	8.9%	136	10	7.4%	0.35
No Prior contraception	131	3	2.3%	135	5	3.7%	136	6	4.4	0.70
Fetal loss < 20 Weeks gestation	98	16	16.3%	95	23	24.2%	95	22	23.2%	0.35
Stillbirth	98	2	2.0%	95	9	9.5%	95	4	4.2%	0.07
Plan future children	131	85	64.9%	135	96	71.1%	136	106	77.9%	0.06
Age lastborn child	92	3.1	(2.9)	87	3.6	(3.1)	89	3.0	(2.4)	0.27
No alcohol	131	71	54.2%	135	78	57.8%	135	66	48.9%	0.34
Non smoker	131	119	90.8%	135	116	85.9%	135	117	86.7%	0.42
Has mobile phone	131	129	98.5%	135	130	96.3%	135	131	97.0%	0.64

Table 1 Baseline variables, expressed as mean values (standard deviation) or number, %

 Table 2
 Hormonal results expressed as median (interquartile range, IQR) values

Treatment	n	Baseline median (IQR)	n	6 months median (IQR)	n	Median change (6 m vs baseline, IQR)	<i>p</i> -value ^a
Testosterone, nmol/L							
DMPA	131	0.82 (0.56, 1.2)	132	0.68 (0.5, 0.88)	131	-0.16 (-0.35, 0.04)	< 0.001
IUD	135	0.9 (0.66, 1.18)	135	0.86 (0.68, 1.11)	135	0.01 (-0.19, 0.14)	0.337
Implant	135	0.87 (0.64, 1.26)	135	0.66 (0.51, 0.85)	133	-0.23 (-0.48, -0.03)	< 0.001
Sex Hormone Binding Globulin, nmol/L							
DMPA	131	52.4 (38.4, 74.1)	132	40.65 (31.55–53.9)	131	-9.6 (-21.8, 0.4)	< 0.001
IUD	135	50.5 (34.7, 76)	135	49.1 (36.2–71.1)	135	3 (–13, 15.5)	0.618
Implant	132	55.75 (34.45, 81.9)	134	23.35 (18.1–33.6)	132	-27.2 (-45.85, -12.05)	< 0.001
Free testosterone, pmol/L							
DMPA	131	10 (7, 16)	131	11 (8, 15)	131	0 (-3, 3)	0.516
IUD	135	12 (9, 16)	135	12 (9, 17)	135	0 (-4, 2)	0.225
Implant	136	11 (8, 16)	132	14 (10, 18.5)	132	3 (-1.5, 6)	< 0.001

^a p-value from Wilcoxon signed-rank test for difference between baseline and six-month values. DMPA, Depot medroxyprogesterone acetate, IUD Copper intrauterine device, Implant – levonorgestrel implant

(mean percentage differences 28.35 and -22.98, p < 0.001 and p = 0.001 respectively) than that for the IUD (0.86 (0.68, 1.11) nmol/L) which was not significantly changed from baseline.

Sex hormone binding globulin (SHBG)

Baseline median SHBG levels were similar between the three groups (DMPA 52.4 (38.4, 74.1), IUD 50.5 (34.7, 76) and implant 55.75 (34.45, 81.9) nmol/L). At 6 months, median SHBG levels in the IUD group were unchanged

Treatment	Baseline mean % difference (CI)	p-value	6 months mean % difference (Cl)	p-value	Change mean % difference (CI)	p-value
Testosterone						
IUD vs DMPA	7.32 (–3.7, 19.6)	0.202	28.35 (15.19, 43.02)	< 0.001	19.60 (7.24, 33.39)	0.001
Implant vs DMPA	10.09 (–1.21, 22.68)	0.082	-1.14 (-11.27, 10.15)	0.836	-10.20 (-19.49, 0.18)	0.054
Implant vs IUD	2.58 (–7.88, 14.22)	0.642	-22.98 (-14.23, -30.83)	< 0.001	-24.91 (-32.63, -16.31)	<0.001
Sex Hormone Binding Globulin						
IUD vs DMPA	-7.73 (-19.23, 5.41)	0.236	21.19 (6.10, 38.42)	0.005	31.34 (15.84, 48.91)	<0.001
Implant vs DMPA	0.37 (–12.19, 14.73)	0.957	-39.45 (-30.82, -47)	< 0.001	39.67 (46.82,31.56)	< 0.001
Implant vs IUD	8.78 (–4.75, 24.23)	0.214	-50.04 (-42.96, -56.23)	< 0.001	-54.07 (-59.48, -47.94)	< 0.001
Free testosterone						
IUD vs DMPA	14.91 (1.37, 30.26)	0.030	13.53 (0.15, 28.7)	0.047	-1.20 (-11.87, 10.77)	0.836
Implant vs DMPA	7.70 (–4.97, 22.06)	0.245	29.60 (14.28, 46.97)	< 0.001	20.34 (7.29, 34.97)	0.002
Implant vs IUD	6.28 (17.22, 6.12)	0.306	14.15 (0.76, 29.33)	0.038	21.80 (8.68, 36.50)	0.001

Table 3 Pairwise comparisons of hormonal results between allocated methods expressed as mean percentage differences (95% confidence interval (CI))

(49.1 (36.2–71.1) nmol/L). Levels for DMPA and the implant were significantly reduced from baseline. That for DMPA (40.65 (31.55–53.9) nmol/L) was significantly lower than the IUD (mean percentage difference 21.19, p = 0.005). That for the implant (23.35 (18.1–33.6) nmol/L) was significantly lower than both the IUD (mean percentage difference – 50.04, p < 0.001) and DMPA (mean percentage difference – 39.45, p < 0.001).

Free testosterone

Baseline median free testosterone levels were similar between the three groups (DMPA 10 (7, 16), IUD 12 (9, 16) and implant 11 (8, 16) pmol/L). At 6 months, median free testosterone levels for DMPA (11 (8, 15)) was lower (mean percentage difference 13.53, p = 0.047) and the implant (14 (10, 18.5) higher (mean percentage difference 14.15, p = 0.038) than the IUD (12 (9, 17). The value for the implant was higher than for DMPA (mean percentage difference 29.60, p < 0.001).

Discussion

This is to our knowledge the first randomized trial to quantify the differences in levels of testosterone, SHBG and free testosterone in young participants randomly allocated to receive DMPA, the LNG implant, or nonhormonal contraception (the IUD). Total testosterone levels of those allocated to DMPA and to LNG implant were significantly lower than for the IUD after 6 months. The striking finding was the significantly greater reduction in SHBG with the implant than with DMPA, which resulted in the free testosterone being 29% higher with the implant than with DMPA. This large difference may have meaningful consequences in terms of side-effects such as acne and hirsutism [20], as well as beneficial effects on sexual function, mood and bone health. The possible association of free testosterone with libido is consistent with the finding in the main Echo Trial [13] of significantly higher reports in the LNG implant than the DMPA group for multiple sex partners, new sex partner, unprotected sex and no condom used for the last sex act. The differences in reported unprotected sex are also consistent with an ancillary study at three of the ECHO sites which found prostate-specific antigen levels in cervical samples to be more frequent in women allocated to the levonorgestrel implant and the Cu IUD than to DMPA [21]. The markedly reduced SHBG with the levonorgestrel implant is in sharp contrast to levonorgestrel containing oral contraception which is associated with increased SHBG [22], probably due to estrogenic stimulation of hepatic SHBG production. The etonogestrel implant has been associated with impaired sexual function ascribed to suppressed estrogen and androgen levels [16]. In contrast, increased free testosterone with the levonorgestrel implant might differentiate user acceptability between the two implant formulations.

Given the findings of this study, it is important not to assume similar side-effect profiles of progestogen contraceptives, but to keep in mind the potential for differential effects of the progestogens on endogenous hormone levels.

The results should be generalisable to similar, low-risk populations.

Limitations

As sampling was at the time of the DMPA administration visit, the DMPA results would reflect the nadir DMPA effect, and the implant results would represent the steady state effect at 6 months. This study could not assess possible post-injection peak effects of DMPA, and thus may have under-estimated differences between the methods.

Spot checks conducted during the main study indicated that a proportion of women had evidence of persistent MPA or norethisterone levels or recent oral contraceptive pill use. Considerable discrepancies between selfreported and biologically confirmed prior contraceptive exposure have been reported in other studies [23]. Use of oral contraception was permitted up to the day preceding enrolment. It is likely that some women in all groups had some testosterone and SHBG suppression at baseline. These effects are likely to be similar between groups due to the stringent randomized methodology. The measured reductions in testosterone and SHBG in the hormonal methods groups may be an under-estimate of the true suppressive effect of the methods. The effects of hormonal contraception may be affected by clinical characteristics such as body mass index and medication. Because the objective of the study was to measure comparative differences between contraceptive methods rather than absolute levels, and because stringent randomization and large sample size would balance such clinical characteristics between groups, as confirmed by the similar baseline characteristics, it was not necessary to adjust for confounding clinical characteristics.

Conclusions

We have compared testosterone, SHBG and free testosterone levels between participants allocated to three contraceptive methods in the context of a robust randomized trial. The significantly higher levels of free testosterone with the LNG implant than with DMPA-IM may have important clinical implications in terms of differential physiological, psychological and behavioural effects, sideeffects and acceptability of these methods, and the tailoring of method choice according to user clinical profiles.

Abbreviations

 ECHO
 Evidence for contraceptive options and HIV Outcomes

 DMPA-IM
 Depot medroxyprogesterone acetate intramuscular

 IUD
 Intrauterine device

 LNG
 Levonorgestrel

Acknowledgements

We thank the participants who participated in this study, the ECHO Study management and collaborators, the staff of the Effective Care Research Unit and Cecilia Makiwane Hospital, the Hospital management, the members of our Community Advisory Board, the Eastern Cape Department of Health and the University of the Witwatersrand Committee for Research on Human Subjects.

Authors' contributions

GJH conceived the study and wrote the first drafts of the protocol and the paper. MS-M and JB gave substantive input into the study design and conduct of the study. YB assisted with data management and analysis. GJH, MS-M, JB, YB and CM contributed to and approved the final manuscript.

Funding

This work and the Evidence for Contraceptive Options and HIV Outcomes (ECHO) Study were made possible by the combined generous support of the Bill & Melinda Gates Foundation (Grant OPP1032115), the American people through the United States Agency for International Development (Grant AID-OAA-A-15-00045), the Swedish International Development Cooperation Agency (Grant 2017/762965-0), the South Africa Medical Research Council, and the United Nations Population Fund. Contraceptive supplies were donated by the Government of South Africa and US Agency for International Development. The contents of this paper are solely the responsibility of the authors and do not necessarily reflect the views, decisions or policies of the institutions with which they are affiliated, the ECHO trial funders, or the supporting governments. The South African Medical Research Council (SAMRC) under the SAMRC Research Capacity Development Initiative (RCDI) Programme funded this ancillary study.

Availability of data and materials

The data that support the findings of this ancillary study of the ECHO Study are available from the corresponding author upon reasonable request. Access will be granted if the concept is evaluated to have scientific merit and if sufficient data protections are in place. As of the time of publication, data access applications are in process with the governing institutional review boards of the ECHO Study to make de-identified data from the primary ECHO dataset publicly available.

Declarations

Ethics approval and consent to participate

The original ECHO Trial was approved by ethics committees at all the participating sites [13]. All participants gave written informed consent to participate. The application for additional tests on stored biological samples was approved by the Human Research Ethics Committee, University of the Witwatersrand on 6 May 2019, reference 141112. The research was conducted in accordance with the Declaration of Helsinki.

Consent for publication

Not applicable.

Competing interests

The authors declare no competing interests.

Author details

¹Department of Obstetrics and Gynaecology, University of Botswana, Notwane Rd, Gaborone, Botswana. ²Effective Care Research Unit, University of the Witwatersrand, Johannesburg, South Africa. ³Effective Care Research Unit, Walter Sisulu University, East London, South Africa. ⁴Effective Care Research Unit, Eastern Cape Department of Health, Bisho, South Africa. ⁵Effective Care Research Unit, University of Fort Hare, Church St, East London, South Africa. ⁶Biostatistics Research Unit, South African Medical Research Council, Durban, South Africa. ⁷Botswana Harvard AIDS Institute Partnership, Gaborone, Botswana. ⁸MRC Centre for Reproductive Health, University of Edinburgh, Edinburgh, UK.

Received: 7 March 2022 Accepted: 22 February 2024 Published online: 08 March 2024

References

- 1. Bosanac SS, Trivedi M, Clark AK, Sivamani RK, Larsen LN. Progestins and acne vulgaris: a review. Dermatol Online J. 2018;24(5):13030.
- Jones EE. Androgenic effects of oral contraceptives: implications for patient compliance. Am J Med. 1995;98(1A):1165–95.
- De Leo V, Musacchio MC, Cappelli V, Piomboni P, Morgante G. Hormonal contraceptives: pharmacology tailored to women's health. Hum Reprod Update. 2016;22(5):634–46.
- Bancroft J, Sartorius N. The effects of oral contraceptives on well-being and sexuality. Oxf Rev Reprod Biol. 1990;12:57–92.
- Evans SF, Kwok Y, Solterbeck A, Pyragius C, Hull ML, Hutchinson MR, et al. The relationship between androgens and days per month of period pain, pelvic pain, headache, and TLR4 responsiveness of peripheral blood mononuclear cells in young women with Dysmenorrhoea. J Pain Res. 2021;14:585–99.
- van Lunsen RHW, Zimmerman Y, Coelingh Bennink HJT, Termeer HMM, Appels N, Fauser BCJM, et al. Maintaining physiologic testosterone levels during combined oral contraceptives by adding dehydroepiandrosterone: II. Effects on sexual function. A phase II randomized, double-blind, placebo-controlled study. Contra. 2018;98(1):56–62.
- Vegunta S, Kling JM, Kapoor E. J androgen therapy in women. Womens Health (Larchmt). 2020;29(1):57–64.
- Coelingh Bennink HJT, Zimmerman Y, Laan E, Termeer HMM, Appels N, Albert A, et al. Maintaining physiological testosterone levels by adding dehydroepiandrosterone to combined oral contraceptives: I. Endoc Eff Contra. 2017;96(5):322–32.
- Hofmeyr GJ, Singata-Madliki M, Lawrie TA, Temmerman M. Hypothesis: amenorrhea-inducing contraception may reduce HIV acquisition risk. Contra. 2014;90(6):615–6.
- 10. Hofmeyr GJ, Singata M, Lawrie TA, Temmerman M. Interpretation, communication, and mechanisms of associations between injectable contraception and HIV risk. Lancet HIV. 2015;2(9):e365.
- 11. Hofmeyr GJ, Singata-Madliki M, Lawrie TA, Bergel E, Temmerman M. Effects of the copper intrauterine device versus injectable progestin contraception on pregnancy rates and method discontinuation among women attending termination of pregnancy services in South Africa: a pragmatic randomized controlled trial. Reprod Health. 2016;13(1):42.
- Hofmeyr GJ, Singata-Madliki M, Lawrie TA, Bergel E, Temmerman M. Effects of injectable progestogen contraception versus the copper intrauterine device on HIV acquisition: sub-study of a pragmatic randomised controlled trial. J Fam Plann Reprod Health Care. 2017;43(3):175–80.
- Ahmed K, Baeten JM, Beksinska M, Bekker LG, Bukusi EA, Donnell D, Gichangi PB, Heller KB, Hofmeyr GJ, Justman J, Kasaro MP. The evidence for contraceptive options and HIV outcomes (ECHO) trial consortium. HIV incidence among women using intramuscular depot medroxyprogesterone acetate, a copper intrauterine device, or a levonorgestrel implant for contraception: a randomized, open-label clinical trial. Lancet. 2019;394:303–13.
- Segall-Gutierrez P, Du J, Niu C, Ge M, Tilley I, Mizraji K, et al. Effect of subcutaneous depot-medroxyprogesterone acetate (DMPA-SC) on serum androgen markers in normal-weight, obese, and extremely obese women. Contra. 2012;86(6):739–4.
- Schaffir JA, Isley MM, Woodward M. Oral contraceptives vs injectable progestin in their effect on sexual behavior. Am J Obstet Gynecol. 2010;203(6):545.e1–5.
- Moreira IFA, Bianchini MP, Moreira GRC, Almeida AM, Rezende BA. Sexual function and metabolic/hormonal changes in women using long-term hormonal and non-hormonal contraceptives: a pilot study. BMC Womens Health. 2020;20(1):240.
- Hofmeyr GJ, Morrison CS, Baeten JM, Chipato T, Donnell D, Gichangi P, et al. Rationale and design of a multi-center, open-label, randomised clinical trial comparing HIV incidence and contraceptive benefits in women using three commonly-used contraceptive methods (the ECHO

study). Gates Open Res. 2017;29(1):17. https://doi.org/10.12688/gates openres.12775.1.

- Vermeulen A, Verdonck L, Kaufman JM. A critical evaluation of simple methods for the estimation of free testosterone in serum. J Clin Endocrinol Metab. 1999;84(10):3666–72. https://doi.org/10.1210/jcem.84.10. 6079. PMID: 10523012
- Bui HN, Sluss PM, Blincko S, Knol DL, Blankenstein MA, Heijboer AC. Dynamics of serum testosterone during the menstrual cycle evaluated by daily measurements with an ID-LC-MS/MS method and a 2nd generation automated immunoassay. Steroids. 2013;78(1):96–101.
- 20. Williams NM, Randolph M, Rajabi-Estarabadi A, Keri J, Tosti A. Hormonal contraceptives and dermatology. Am J Clin Dermatol. 2020;22:69–80. https://doi.org/10.1007/s40257-020-00557-5.
- Deese J, Lien Chen P, Gao X, Heffron R, Miller A, Steiner M, et al. Postrandomization differences in condomless vaginal sex among women randomized to DMPA-IM, copper IUD and levonorgestrel implant in the ECHO trial. J Int AIDS Soc. 2021;24(S1):e25659.
- Zhao XJ, Shao GX, Zhang J, Sang GW. Effects of steroid contraceptive on serum levels of sex hormone binding globulin in women. Shengzhi Yu Biyun. 1992;12(4):13–7.
- Achilles SL, Mhlanga FG, Musara P, Poloyac SM, Chirenje ZM, Hillier SL. Misreporting of contraceptive hormone use in clinical research participants. Contra. 2018;97(4):346–53.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.